Contents lists available at ScienceDirect





Ecological Economics

journal homepage: www.elsevier.com/locate/ecolecon

Ecological economics and degrowth: Proposing a future research agenda from the margins



Ksenija Hanaček^{a,*}, Brototi Roy^a, Sofia Avila^a, Giorgos Kallis^{a,b,c}

^a ICTA-UAB, Institut de Ciència i Tecnologia Ambientals, Universitat Autònoma de Barcelona, Research and Degrowth, Spain

^b Departament de Geografia, Universitat Autònoma de Barcelona, Spain

^c ICREA, Pg. Lluís Companys 23, 08010, Barcelona, Spain

ARTICLE INFO

Keywords: Ecological economics Degrowth Global South Feminism Gender

ABSTRACT

Research by ecological economists on degrowth is a flourishing field. Existing research has focused on limits to (green) growth and on economic alternatives for prospering without growth. Future research, we argue here, should pay more attention to, and be written, from the "margins" – that is from the point of view of those marginalized in the growth economy. We conduct a comprehensive systematic review of the prevalent themes in the existing literature on the ecological economics of degrowth, and its engagements with North-South relations and gender issues. The analysis identifies seven research areas where ecological economics can better integrate these matters, namely: the study of post-growth policies for the Global South; the unequal exchanges that sustain an imperial mode of living; the deconstruction of ecological economic concepts that reproduce problematic Western or gendered assumptions; the study of the clash of metabolisms in peripheries of the Global South; the metabolism of care-work in growth economies; the leading role of women in ecological distribution conflicts, and the reproduction of gender inequalities in alternative post-growth spaces. We propose that ecological economics should welcome more contributions from critical feminist scholarship and scholars from the Global South.

1. Introduction

Ecological Economics (EE) and Degrowth are both relatively new research paradigms which share origins and common paths (Røpke, 2004; Weiss and Cattaneo, 2017). EE can be defined as the pre-analytic vision of the economy as a physical system embedded in the finite, complex planetary ecosystem with the explicitly normative goals of achieving ecological sustainability and just distribution (Daly, 2013). EE developed as an interdisciplinary field for understanding and managing the connections between the economy, ecosystems and the social structures that sustain human well-being (Zografos and Howarth, 2010). A social strand of EE research, however, has developed a more critical, political-economic approach which emphasises distributional, institutional and power issues (Kallis et al., 2009; Martinez-Alier, 2015; Spash, 2011). This social EE has more recently encountered the idea of degrowth.

Degrowth was born as a proposal for radical change, and is seen by its advocates as an explicit attempt to re-politicize the environmental question (D'Alisa et al., 2015; Kallis et al., 2012; Martínez-Alier et al., 2010). Degrowth emerged as an activist slogan in the 2000s, first in France and then the rest of Europe, branching into a transdisciplinary academic paradigm, largely due to the biennial international conferences since 2008 (Demaria et al., 2013; Kallis et al., 2018; Sekulova et al., 2013). The slogan itself can be traced back to the title in French *Demain la décroissance: entropie-écologie-économie (Tomorrow's Degrowth: Entropy-Ecology-Economy)*, a translated collection of essays of Georgescu-Roegen (1979) edited by Jacques Grinevald and Ivo Rens (Demaria et al., 2013). Both Degrowth and EE recognize the need to limit the scale of the economy within planetary boundaries and understand the economy as fundamentally embedded in social and ecological systems.

However, the case against growth in Degrowth literature rests not only on ecological limits, but also self-limitation and voluntary simplicity, equality, global social justice and an end to the exploitative relations of capitalism (Kallis et al., 2018). In recent years, a research agenda on degrowth has emerged within EE (Kallis et al., 2012). The agenda has made important contributions to understanding the limits of green growth and exploring alternative economic policies and grassroots arrangements for living without growth (Kallis et al., 2018).

The objective of this article is to document gaps in this literature

* Corresponding author.

E-mail address: ksenija.hanacek@gmail.com (K. Hanaček).

https://doi.org/10.1016/j.ecolecon.2019.106495

Received 28 March 2019; Received in revised form 16 August 2019; Accepted 26 September 2019 Available online 11 December 2019

0921-8009/ © 2019 Elsevier B.V. All rights reserved.

and propose directions to fertile areas for further research. The intention is to cultivate a research agenda 'from the margins' – that is from the point of view of groups exploited within the growth economy. The article focuses, in particular, on communities and people in the Global South; the unequal exchanges between North and South; and unequal gender relations in production and reproduction.

As this study demonstrates, both within Degrowth and EE literatures there is a lot of lip service paid to the need to address global injustices and to relate concepts such as degrowth or the steady-state to the Global South. Furthermore, there is a need to pay more attention to unpaid care work, and the exploitation of women under unfair economic and ecological arrangements. However, few research studies are actually carried out in this regard (Dengler and Seebacher, 2019). In this article, thus, concrete proposals for prospective research on gender and the Global South are outlined.

The paper is structured as follows. Section 2 explains the methods we used for a systematic analysis of the literature, and section 3 provides results by using network diagrams which depict the current state of Degrowth literature within EE, and the treatment of gender and North-South issues within it. Network analysis helps us identify the core themes and narratives in the existing literature. Our analysis confirms that the questions posed by feminist and Southern scholars are still treated lightly within EE. Section 4 reviews qualitatively relevant studies beyond the immediate field of EE of Degrowth and provides seven relevant research areas or topics (four for Global South and three for feminist ecological economics). Section 5 concludes.

2. Methodology

This study is based on a qualitative and in-depth systematic literature review. The method represents a comprehensive approach that allows identifying, interpreting and evaluating available literature (Gibbs, 2004). In this case, three literatures were evaluated: the EE of Degrowth, debates about the Global South within it, and feminist ecological economics (FEE). Fig. 1 illustrates visually the methodological design.

By using the advanced search tool in the Scopus database (22/6/19), the literatures were identified as follows:

a)

- a) "Ecological Economics" AND "Degrowth" AND "Global South". Using the advanced search tool while limiting literature to the subject area of environmental science (SUBJAREA- ENVI), 314 results were obtained spanning from 2010-2019. Each study was skimmed and Global South discourses were identified in 109 papers, which included synonyms such as "less developed countries", "oppressed countries", "South-North", "indigenous communities", "periphery", "migrants", "ethnic minorities", "outskirts", "urban and rural poor", etc. Only 4 articles within this set addressed gender inequalities. The 109 articles were selected for the analysis spanning the 2010–2019 period (Appendix A).
- b) "Feminist" AND "Ecological Economics" AND "Gender" and "Women". This resulted in 3210 articles spanning from 1988-2019. Again, the obtained result was limited to environmental science subarea (SUBJAREA- ENVI) which rendered 395 results. Then, a third criteria was applied to only include critical feminist approaches, as considered to be more closely related with the Degrowth perspective. The aim of critical feminist approaches is to explicitly deconstruct economic relations in terms of power domination and privilege, focusing on the social construction of exploited environments, women, and workers/peasants (class) (MacGregor, 2017; Perkins, 1997). In total, 108 articles were selected spanning the 2010–2019 period (Appendix B).

The analysis included only articles, conference papers, and book chapters available in English, excluded duplicate articles as well as articles that after reading were found to be addressing women and nature, but with no links to broadly-defined economic issues.

The 217 articles were imported into the qualitative analysis



Fig. 1. Methodological review approach for identifying, interpreting and evaluating literature on Ecological Economics and Degrowth, Global South, and feminism/ gender/women for future research agenda.

K. Hanaček, et al.

Table 1

General categories (parent nodes)	N of subcategories (child nodes)	N of frequency of co-occurrence between i) general categories and subcategories; ii) between different subcategories
Ecological Economics and Degrowth and Global South	147	2929
Feminist Ecological Economics	165	5813

software NVivo (QSR, version 11.0) where categories of interest were coded. Following Siccama and Penna (2008) the coding for each category was structured hierarchically. The general codes (so-called "parent nodes") that were at the top of the hierarchy include core search terms: "Ecological economics, Degrowth and Global South", "Feminist Ecological Economics" (Table 1).

Each article was assigned one or more of the core search terms. Different terms and concepts as subcategories (so-called "child nodes") were coded paragraph by paragraph. The choice of terms and concepts was guided by issues affecting different geographical and social groups in the growth economy.

For example, a paper on the economics of degrowth (Ref. 1) was examined, in which the terms "exploitation' and "debt" appeared in a paragraph. The terms were classified under the parent node "ecological economics, degrowth". The software registered the subcategories "exploitation" and "debt" below the parent node (Table 1). If the term "debt" appeared three times within the paper, it was coded three times. The software counted then one connection between "exploitation" and ecological economics of degrowth and three connections between "debt" and ecological economics of degrowth (ref 1; "exploitation" = 1; "debt" = 3).

Whether there was a connection between subcategories "debt" and "exploitation" depended on whether the two terms appeared "together" in the same paragraph; where the author of the reference makes an argument between the two terms. If the terms appeared "together" more than once in a paper, then there was more than one connection between them (Ref 1; "debt" X "exploitation" = 7); but there could also be zero connections (e.g. Ref 2; "debt" X "exploitation" = 0). The frequency of connections between terms (codes) gave a sense of whether they were related or not in the literature.

The connection of terms was performed by using a descriptive statistics which allowed characterization of the selected data based on its attributes (as in: Siccama and Penna, 2008). Within descriptive statistics, frequency of co-occurrence showed how often attributes (terms) appeared "together". This allowed deciphering patterns of interaction between terms in the literature, helping reconstruct broader narratives in a more objective, and measurable way. For the representation of the frequency of co-occurrences between terms (attributes/codes), networks were designed in Gephi software (version 0.9.2). A complete methodological outline is presented in Fig. 2.

3. Network analysis of the literature

3.1. Ecological economics of degrowth: studies on Global South and gender

In this section we assess the current state of the peer-reviewed literature on the ecological economics of degrowth and how it fares in relation to issues connected to the Global South and gender.

Fig. 3 shows the geographical distribution of studies on the ecological economics of degrowth and the Global South (109 in total). Each paper was examined separately and the country or countries were the research was undertaken was registered. For conceptual papers with no empirical studies, "no country" was assigned.

The map shows that most of the empirical studies focus on India, China, and countries of South East Asia. Studies on Latin America followed in number of publications with most studies on Ecuador,

Colombia and Mexico. Studies in Brazil, Argentina and Peru were covered to a lesser extent. In the South of Europe, historically considered as a relatively "less developed" region within the continent, we found ongoing debates in Spain and Italy. "South within the North" debates were also identified in Germany and France. As the map furthermore shows, there was hardly any research focusing on African countries, a continent with tremendous marginalisation and oppression. Exceptions were South Africa, Nigeria, and Cameroon.

By further analyzing this same set of 109 articles other relevant patterns were found. Fig. 4 shows that studies on the EE of Degrowth addressing the specific issues of the Global South have been increasing since 2010. In contrast, issues related to feminisms, gender and women, were still poorly considered, as only 4 articles out of the 109 publications engaged with these matters. This illustrates that the study of gender relations is still a marginal topic within the literature on the Ecological Economics of Degrowth.

When analysing feminist scholarship separately, another set of 108 publications emerged. Fig. 5 shows that many of the studies on gender inequalities were conducted in the United States and Canada. Studies on inequalities and struggles of women in India, Mexico, Brazil, Chile, Bolivia and China followed in number of publications. In the case of Africa, most studies were found in Kenya, Ghana, South Africa and Uganda. Yet, a large area of Africa remains under-investigated. Furthermore, hardly any work was found on issues of gender inequalities in Eastern Europe. In addition to this, Fig. 6 shows there was an increase in critical feminist scholarship literature in 2018, according to the number of publications per year.

3.2. Ecological economics of degrowth: Global South and gender narratives

In this section, an overview and the main narratives contained in the literature on the EE of Degrowth is presented. In Fig. 7 the size of a circle indicates the frequency of appearance of the term at stake, and the width of the connecting lines indicates the frequency of connections between different themes or concepts - meaning the rate at which themes and concepts appear together in the selected papers. Proximity between themes indicates that they are more frequently associated. Density of the connecting lines within the network indicates higher frequency of debates within the literature.

The most prominent themes are the central nodes, in this literature limits to economic growth; environmental impact, policy institutions and law; global economic growth as well as distribution conflicts. These are the themes most interrelated with the rest of the themes. The closeness of nodes to the centre signify their relevance in the debate; less frequent discussions are represented by nodes situated in the periphery of the network. Some of the smaller size themes are situated closer to the central node. This is because their frequency of connection with the central themes within the reviewed literature is high (e.g. crisis and transition).

From the network diagrams and informed by our reading of the literature, we can start constructing some narratives. According to Fig. 7., the literature on EE and degrowth is focussed on limits to growth and energy questioning global economic growth, due to its huge dependency on natural resources and the environmental impacts it has, and it sees such growth as part of current capitalistic and neo-liberal systems. We further see that policies, institutions and laws for sustainability as well



Fig. 2. Methodological outline: data gathering, organising, and analysis.

as alternative economy and practices are a main focus of research in the literature. These code themes are also frequently related to notions of reduced production and reduced consumption and working hours for transformative development and increasing welfare and well-being, which is coherent with the approach of ecological economists working on degrowth.

To a lesser extent, articles question development as an ideology imposed by current capitalist and neoliberalist economic system especially by the Global North; as well as the need for politicization and politics of transformative development, physical and ecological limits, and inequalities as a consequence of global economic growth. Discussions of privilege, valuation languages, class, gender, history and culture are important, but relatively less addressed within the reviewed literature. Although less covered, gender issues within the EE of degrowth literature relate to arguments such as segregation, work, and place, territory and space.

Fig. 7 furthermore depicts interrelated discourses within the subset of articles on the ecological economics of degrowth that cover the Global South. Closest to the Global South theme are the most recurrent concepts: *global economic growth, inequalities and resistance to* these situations. The analysis furthermore confirms that the degrowth discourse about the Global South is still at an early stage, focussed on making the point that the material extraction (mostly for *energy* production) that fuels global economic growth has negative social and environmental impacts in the countries of the South.

A second category includes terms that characterize the state of North-South dynamics such as *power relation, marginalization, extraction, exploitation, expansion, development* and *capitalism,* pointing to less studied phenomena that ecological economists might want to explore further. Then, *justice* and social-environmental *conflicts* in terms of



Fig. 3. Number of articles on EE, Degrowth, and the Global South based on the countries of research.



Fig. 4. Number of articles on Ecological Economics and Degrowth studied per year, of which number of articles that address the Global South and feminism/gender/ women perspectives that are included in the analysis (109 studies in total).

environmental *distribution issues* and *El Buen Vivir* are identified frequently as an alternative to natural resource *extraction, exploitation and expansion*.

This is followed by the third category of concepts that refer to: i) dynamics such as power of knowledge, history, westernization, production, class and race, ii) cultural differences, debt, livelihoods and survival, iii) possible paths to transform unjust relations such as transformative development, sustainability, alternative economies and practices; and reduced consumption as either a choice or a condition (poverty).

The limits to growth argument remains central in this literature, pointing to the fact that arguments about global injustice are framed in relation to environmental limits of *global economic growth*. The analysis finds a common narrative in the degrowth literature, whereby global economic growth is facing limits, and therefore the North should limit itself so that the South has more space to grow. It is also observed how *culture, identity and traditions* still remain poorly addressed within the EE literature, as well as history, coloniality, and ecological debt.

For the last part of the analysis (Fig. 8) we look further into the themes that feature prominently in the 108 studies of feminist EE. The main debates in feminist EE are around the relationship of *women* and *nature and environment*, found as one would expect at the centre of the network. Proximity to the central nodes demonstrates how contributions touch upon the fact that *gender issues* related to *care and reproduction* work carried out by *women* are *invisible and ignored* and this is further related to *struggle, resistance and empowerment* through *feminist movements*.

Global South debates in the feminist EE literature explain issues of *marginalisation* and *inequalities* as a result of development *policies* that result in further exploitation and feed poverty, rather than wealth and well-being. Further, North-South relations are described through *power relations*, *technocracy* or green "solutions" in environmental management, *dispossessions* and *injustices*. The literature also points to exclusion



Fig. 5. Number of articles on critical feminist scholarship based on the countries of research.



Fig. 6. Number of publications per year on critical feminist scholarship included in the analysis (108 studies in total).

of women from environmental decision-making; arguing how *sustainability* should be based on a *participation* of all members of the *society*, taking into account their *livelihoods and existence*.

Furthermore, studies link social degradation and environmental degradation with the struggles, resistances and empowerment of women against oppressive structures for their livelihoods and existence, involving often the community in its entirety. Feminist debates in Ecological Economics engage with unpaid labour, the importance of different cultures and identities, especially in racialised settings; and divisions along the lines of class (rural and indigenous communities) and the problems with dualisms, that must be deconstructed in the pursuit of democracy and integrity.

The high number of theme codes (Fig. 8) indicates a fragmented literature with many small debates, many of them yet to be taken up, such as problematic discourses about *biological differences* and *inferiority* between genders. The periphery of the network indicates the least explored topics, which are in no way unimportant, with a possible discourse around *growth-oriented exploitation* resulting in *commodification* and *injustice* due to *patriarchal power relations and sexism, racism, violence, war, genocide,* and *displacements.* The issue of *class,* (along with that of *race*) is present in this feminist literature pointing to an intersectional literature, though less represented within the broader EE of degrowth literature.

4. Discussion and prospects for further research

There is an increasing interest in expanding current EE of Degrowth research agenda (Kallis et al., 2018). Yet, as this and previous findings suggest, Global South and feminist approaches are far from studied thoroughly (Dengler and Seebacher, 2019). This article complements previous studies with a more qualitative and extended approach of the ecological economics of degrowth literature, by analysing and considering broader scholarships that are relevant to the study of degrowth vis-à-vis the Global South and gender relations. The following section characterizes some of the key points developed by the existing literature presented in section 3 and then identifies new relevant themes that should be explored further. 4.1. Ecological economics of degrowth and the global south: emerging debates

Section 3 has shown that current scholarship on the ecological economics of degrowth continues to dabble in well-established narratives on the Global South and gender issues. These include preliminary links between global economic growth and resource extraction in the South, limits to growth and the need for North-South redistribution, exploitation of the unpaid care work of women, and links between women and environmental justice mobilizations. As observed, these matters are certainly relevant and require further conceptualization and geographical coverage. In parallel, however, this article argues that other peripheral themes and inter-relations need to be more systematically explored. This includes the interactions between race, class, and gender; violence, control, emotions and feelings.

Ongoing scholar debates prove an increasing interest in exploring possible alliances, tensions and contradictions between degrowth and other intellectual, social and political movements questioning capitalism and modernity from the lens of the Global South (Rangan and Sivaramakrishnan, 2001; Tortosa, 2016). Examples include dialogues and tensions between degrowth and environmental justice (Martinez-Alier, 2014; Martinez-Alier et al., 2016; Rodríguez-Labajos et al., 2019); degrowth and neoextractivism (Svampa, 2015); degrowth and "*El Buen Vivir*" (Acosta, 2015; Escobar, 2015; Gudynas, 2014, 2009), or degrowth and the pluriverse with a focus on alternative cultural, economic, social, political and ecological visions and practices (Demaria and Kothari, 2017).

A new agenda for EE could certainly inform and benefit from the current political momentum, where dialogue between degrowth and intellectual and social movements from the Global South is increasing (Martinez-Alier, 2014; Martinez-Alier et al., 2016). As this study argues, a new research agenda should address colonial differences, geopolitics of knowledge, heteropatriarchy and relations of exploitation in the modern capitalist world-system (Grosfoguel, 2011, 2002). This new agenda *"from the margins"* should thus include different theoretical and epistemological traditions (see: De Sousa Santos, 2016; Zibechi, 2006) that go beyond Eurocentric liberal discourses, and engage with the pluriverse of alternatives (Demaria and Kothari, 2017).

While 3 our review of the literature only focused on the most recurrent subjects of study for the Global South, a renewed agenda on the ecological economics of degrowth necessarily involves further



Fig. 7. Themes and connections in the ecological economics of degrowth. The central, the bigger, and the darker the theme circle, the higher is its frequency of the debate within the literature. The closer the circles, the more frequent are their interrelations. The closer the circles, the more frequent are their interrelations. Density and centrality of themes' connection lines indicate their higher frequency of debate within the literature.

engagements with perspectives *from* the Global South. Going beyond Eurocentric approaches has been an ambition present in the history and evolution of Ecological Economics. Contributions from Global South scholars and activists have played a key role in developing some research lines of the field, including theories of ecological unequal exchange, and plural valuations. Such intellectual endeavours have either emerged or been inspired by social movements, which ultimately give force to many traditions on critical studies in the Global South (Svampa, 2016).

The notion of "ecological debt", has its roots in Latin American civil society organizations during the 1990s, and since then has largely influenced the EE discipline (Martínez-Alier, 2002a, 2002b). These include research on quantifying the ecological debt (Matthews, 2016; Neumayer, 2000; Srinivasan et al., 2008) as well as discussion on its content, potential incidence and possible pitfalls (Goeminne and Paredis, 2010; Martínez-Alier, 2002b). Of particular relevance are those works lead by Global South researchers on specific case studies of ecological debt, such as Colombia (Borrero Navia, 1994), India (Khatua and Willam, 2006), Africa (Ngosso, 2016) and Latin America in general

(Azamar-Alonso and Carrillo-González, 2017). Regional societies for EE should play a key role in promoting further integration of cultural diversity, plurality, integration for decolonial thinking, and activist-academic collaborations (as in Demmer and Hummel, 2017; Grosfoguel, 2011).

Degrowth researchers have always been clear on claiming multiple ecological and social sources for the concept (Bayon et al., 2010; Demaria et al., 2013; Flipo, 2007). The literature on degrowth, however, remains largely Western (Weiss and Cattaneo, 2017). Normative visions such as democracy, justice, anti-utilitarianism, and wellbeing have intellectual foundations and alternative versions in the Global South (Gerber and Raina, 2018). Following the principle of "deconstructing the social imaginary" (Latouche, 2009), a decolonial theory of degrowth must acknowledge that many post-growth ideas have non-Western roots, and voices of academics and activists from the Global South are as crucial as those critical streams of thought in the North.

For example, Mahatma Gandhi, one of the foremost leaders of the Indian freedom struggle, known for leading a non-violent civil disobedience movement, followed a philosophy of "simple living, high



Fig. 8. Interrelated views on Feminist Ecological Economics. The central, the bigger, and the darker the theme circle, the higher is its frequency of the debate within the literature. The closer the circles, the more frequent are their interrelations. Density and centrality of themes' connection lines indicate their higher frequency of debate within the literature.

thinking". Gandhi, along with his chief economic advisor, *J.* C. Kumarappa believed in a decentralized village economy, calling it the "economy of permanence" (Govindu and Malghan, 2016, 2005). They argued that industrial society would destroy the earth and sought to rebuild the rural economy on ecological lines by working on water conservation, recycling, the replacement of chemical fertilizers by organic manure, and renewal of craft traditions. These ideas resonate strongly with what degrowth propagates.

In a similar vein, Rabindranath Tagore, one of Bengal's foremost poet, writer and philosopher wrote in 1924 that *"It is food which nourishes, not money; it is the fullness of life which makes one happy, not fullness of purse. Multiplying material wealth alone intensifies the inequality between those who are privileged and those who are dispossessed"*. These lines exemplify some of the rich philisophical traditons present in the Global South which degrowth research must acknowledge and learn from.

Indigenous and grassroots movements in Latin America with long traditions on autonomy and "El Buen Vivir" are not just an inspiration for Degrowth scholarship and activism of the North, but most importantly, a source of radical de-colonial thinking in their own right. This goes hand in hand with the important tradition on critiques to development from Latin American scholars (Escobar, 2015; Esteva,

2010; Gudynas and Acosta, 2011) as well as works contributing to debates unfolding within the degrowth community, but with their own narratives and specificities; including key topics such as municipalization (Olmedo, 2010); feminism (Lozano and Lopez, 2013); autonomy (Esteva, 2018) and communality (Guerrero-Osorio, 2018).

While all these contributions certainly challenge the growth-based development model that sustains modern economic relations, not all of these perspectives identify themselves as part of a "degrowth theory". While the latter has been mostly articulated in Western Europe, critical economic thinking across the Global South has its own trajectories and identities. As such, a more systematic attempt in Ecological Economics must be made to look at various alternative economic approaches emerging at different geographies: not only in the North and West, but also in the South and East. Acknowledging the plurality of intellectual traditions as well as the commonalities between them (in terms of ethical values and principles) is a key aspect for a new research agenda in this field.

4.1.1. Research topic 1: economic policies in the Global South

Ecological economists have proposed several degrowth-oriented policies (Cosme et al., 2017), however little work has focussed on the

relevance of such policies outside Europe and North America. How can work sharing (reduction of paid working hours) be implemented, for example, in economies that are not fully industrialized? Is there a place for a carbon tax (or dividend) or a basic income in low-income economies? If not, what policies would work in such contexts, and steer these economies to more sustainable, post-growth trajectories?

A recent example can be found in a publication made by the Latin American Societies of Ecological Economics (Weiss, 2017), in which distribution of wealth, reduction of metabolic intensity of regional economies and urban-rural relations are addressed. This research agenda requires engaging further with the work of critical scholars and activists from the Global South. A good number of articles have been published by Latin American scholars aware of the contradictions of recent developmentalist governments, proposing alternative policies for a post-extractivist and "El Buen Vivir" horizon (e.g. Acosta, 2012; Bertinat, 2013). Examples of progressive environmental policies can also be found in studies on African experiences, particularly for the management and distribution of oil revenue (for case studies in Sao Tome and Chad see: Bryan and Hoffman, 2007; Sovacool et al., 2016). How do these policies work, under what conditions may they be implemented, and what could lead to their success?

4.1.2. Research topic 2: economic growth, development and north-south relations

New research should move beyond the rigid redistributive dichotomy of "Degrowth in the North – growth in the South". Escobar (2015) has noted how such a distinction is problematic in both epistemic and practical terms: first, it overlooks the important intellectual *critiques to development* and experiences *beyond development* unfolding in the Global South. Second, it forecloses spaces for dialogue between different eco-civilizational transitions, which are essential for an effective politics of transformation.

Disengaging North and South trajectories also ignores the co-constitution of historical binaries between modernity, development, and underdevelopment; as well as rich and poor (Avila-Calero and Pérez-Rincón, 2018; Esteva, 2010). This dialectic approach stems from Latin-American Marxist theorists (e.g. García Linera, 2012), who have argued that "underdevelopment" and hence "dependency" is not a national, isolated phenomenon, but the result of the unequal relations established with former metropolitan powers and current centers of global markets. Unequal exchange is expressed in financial, material and political terms. It represents, in turn, the basis and core explanation of the origins and development of capitalism in Europe and the United States (Marini, 1972; Cueva, 1998). North and South are thus intricately linked through global flows of money and resources through patterns of unequal exchange and the uneven power relations that derive from them. Rather than thinking of growth in the North and the South as separate choices or phenomena, a new research agenda needs to advance into more integrated, regional and global understandings of economic interdependencies and relations.

There is a valid reasoning behind the idea "degrowth in the North" to let the South mark its own trajectory (e.g. Kerschner, 2010). However, the postulate is problematic when the South appears as a unified entity, with no plurality in terms of socio-political experiences and metabolic configurations. Countries in the Caribbean, Brazil and Mexico; Vietnam, India and China, each and every one of them demand special empirical attention and analysis. A recent study focused on China, for example, has shown that even with the rapid economic growth experienced over the last decades, low-income people will be still locked in continual or even worsened poverty. To speak of "China" can be then misleading when one deals with such a large nation with diverse internal experiences (Xue et al., 2012).

This new research agenda should be willing to question the usefulness, or at least limits, of the very notions of "North" and "South". While ecological economists have contributed the most in demonstrating the patterns of unequal exchange between economic cores and peripheries (e.g. Hornborg and Martinez-Alier, 2016), the geographical distinction between North and South is becoming challenging to maintain. Socio-environmental asymmetries are not always so neatly categorized. Some scholars talk now of a "North in the South", the rise of very rich and a growing middle class in poorer countries, or of "South in the North", impoverished people in richer countries and sharpening domestic divisions related to class, race and gender inequalities (Acosta, 2015; Escobar, 2015; Gudynas, 2014).

The Global South, thus, should not be understood strictly as a geographical concept, even though the great majority of impoverished or marginalized people live in countries of the Southern hemisphere (de Sousa Santos, 2016). As such, the South is increasingly used as a metaphor for excluded, silenced and marginalized populations that within our current social-economic-political system experience poverty, displacement, environmental pollution and destruction (Escobar, 2015; Gerber and Raina, 2018).

Using this lens of "the South in the North", our review revealed that there are hardly any publications on Eastern Europe – a periphery of Europe or a region within Europe subjugated by core Western countries in its economic, cultural and environmental terms (de Sousa Santos, 2016; Roncevic, 2002). The exceptions are for example the recent work of Domazet and Ančić (2019) and Velicu (2019) published as a part of the special issue on the potential for a theoretical and political alliance between degrowth and environmental justice (Akbulut et al., 2019). A new generation of work in unequal exchange should refine its scale of analysis and capture the new patterns of exchange and exploitation within the so-called Global North; the emergence of new centres of economic power in the South; and the role of a transnational global elite, present both in the northern and the southern hemispheres, which is moving its capital around the world, dictating the terms of exchange between different places.

The concept of an imperial mode of living (Brand and Wissen, 2018) can be an interesting avenue for ecological economists to explore further, as it shifts attention to centres of accumulation, consumption and the extraction of labour and resources from regional or international peripheries. A rich person in Delhi can exemplify an imperial mode of living as much as a rich person in Britain – a mode sustained by the extraction of resources and the accumulation of ecological debt on the poorer peripheries, workers and environments. This concept is particularly interesting from a degrowth perspective as it links unsustainable, high-throughput modes of living by transnational elites and a global middle class, to patterns of unequal exchange, cost-shifting and environmental degradation in the periphery.

The suggested approach could also interlink further with questions over measures of wealth and development that dominates the assessment of living conditions in "the South". This includes a deconstruction of widely accepted opinions about the economies of agrarian scale and efficiency, based on the assumption that technological solutions directly provide stable rates of well-being. In the case of Eastern Europe, for example, studies have shown that Romanian villages with big and high-input farms have higher poverty levels in contrast with small-scale traditional farms. The latter not only have less poverty, but also register higher environmental, economic and socio-cultural well-being.

4.1.3. Research topic 3: decolonizing ecological economic concepts

Post-development scholars have deconstructed the assumptions, colonial relations and violence embedded in supposedly neutral, "scientific" metrics such as the World Bank's poverty indicators, GDP league tables and the like (e.g. Escobar, 2011; McAfee, 1999). These metrics frame alternative, non-Western ways of living as "backward", posing market-driven growth as the only alternative, serving the interests for Northern economies to expand their markets and access to cheap labour and raw materials (Galeano, 2004). International development programs, thus, sustain colonial relations in the post-colonial era (Escobar, 2011; Esteva, 2010).

Ecological economists should push further their critique of growth-

based development and associated framings of poverty. For example, the concept and metrics of wellbeing (O'Neill et al., 2018), largely a UN and World Bank construct (e.g. McAfee, 1999), presupposes a Western view of development, and has little room for considering alternative, indigenous cosmovisions or other diverse ways of living well. This is similar to the prevalent view within ecological economics of seeing ecosystems as providers of services. This notion alienates many of the local people living and working in regions with high bio-cultural diversity (Geijzendorffer et al., 2017). The seemingly innocuous concept of ecosystem services exports with it a whole way of seeing and managing society-nature relations (Kallis et al., 2013). For example, understanding nature only as capital which provides goods and services – is an understanding that is not politically neutral (Hanaček and Rodríguez-Labajos, 2018; Kull et al., 2015).

This calls for research within the EE community following the lead of post-development scholars in documenting how different discourses travel and reveal their untold assumptions and undesired implications. This readiness to address should go as far as including in the critique the very concept of degrowth (Mignolo, 2018). The concept of degrowth creates frictions with environmental justice movements in the Global South which do not necessarily feel comfortable with this framing of their issues or visions. The keys points of contention range from claims of it being too anthropocentric and Eurocentric, to nonacknowledgement of the multiple ways of organizing, understanding and discussing issues affecting groups in different parts of the world, to semantic controversies and the use of detached ideas and approaches (Rodríguez-Labajos et al., 2019). Those developing degrowth theory and advocacy have much to learn by studying how the concept travels (or does not travel) to other places of the world. In turn, this can enrich the degrowth concept and "make it travel better", by recognizing and incorporating in its theoretical corpus ideas that come from scholars from other parts of the world.

4.1.4. Research topic 4: metabolic studies and approaches to transitions

There is an important body of research on the metabolism of advanced industrial systems (e.g. Giampietro et al., 2013). There are also studies on the changing metabolisms of "peripheral" economies in relation to patterns of export-driven growth, and the growth of consumption in the Global North (Gonzalez-Martinez and Schandl, 2008; Pérez-Rincón et al., 2018; Russi et al., 2008; Vallejo, 2010). An interesting development is the linking of this 'metabolic approach' to the question of socio-ecological transitions, and changes in socio-metabolic regimes unfolding throughout human history (Haberl et al., 2010). The analytical emphasis on the transition dynamics from agrarian to industrial regimes across the globe has been certainly informative. and is increasing complemented by the interesting questions around transitions towards alternative socio-metabolic configurations today (e.g. Mingorría et al., 2014; Ravera et al., 2014).

The analysis of the contesting metabolisms of traditional communities and mining economies in Conga in the Andes-Amazon based on Georgescu-Roegen's flow-fund model (Silva-Macher and Farrell, 2014) is an interesting approach that researchers could adopt when studying conflicts between local communities defending their ways of living and the metabolism of the development process. An appreciation of both the environmentalism and the metabolism of the poor (Martinez-Alier, 2015, 2014) has been central for a degrowth and ecological economic research agenda. Metabolic studies can buttress the thesis that often the sustainable systems are subsistence economies, that have low throughput; and not the rich-industrial societies, which despite improvements in efficiency, use much more resources.

4.2. Ecological economics and degrowth engagements with feminist thinking

4.2.1. State of the literature on ecological economics, degrowth and feminism

As the degrowth community seeks to strengthen and radicalise EE, it

is important to incorporate feminist perspectives (Perkins, 2007). Feminist scholarship shows how women lack recognition for their reproductive work. Most women still carry out work in the home; a work that is 'destined' to be unwaged. Ostensibly, the purpose of this unwaged work is to reproduce their own families but ultimately it serves to provide the capitalist class with the labour power the economy requires and with a new generation of workers (Federici, 2018a).

From the feminist environmental governance point of view, women's history of exclusion from public institutions and politics, and the enclosures of rural women from their commons, such as access to domestic fuel and clean water, constrain their effective economic participation. Women also often lead movements against these injustices (Agarwal, 2009; Perkins et al., 2005). Economists tend to underestimate the role of both the environment and the work of women for the sustenance of economic activities. Hence, there is still work to be done to bring feminist and ecological economics closer together (Nelson and Power, 2018).

Feminist theories provide dynamic, inter-disciplinary, communityoriented, and pluralistic approaches (Bauhardt, 2014; Perkins et al., 2005; Spencer et al., 2018). Although both EE and degrowth literatures discuss feminist theories and address the need to incorporate them, both fields still remain gendered in terms of authorship, intellectual history, and content (Dilli et al., 2018; Woolley, 2005).

There is for example very little theoretical or empirical research on the gendered and racial nature of the care economy (Bauhardt, 2014) or the role of women's reproductive labour, slave labour, migrant labour, labour in the Global South or the work of the unemployed and their contribution to the ecological-economy of capitalism (Federici, 2018b). Feminist scholars have shown how societies have commonly been characterized by the subjugation of women to male authority, exclusion of women from environmental governance (e.g. Agarwal, 2010) both with the family and in the community in general; the objectification of women as a form of property; a gendered and racial division of labour in which women of ethnic minorities are confined to activities such as child-raising, performing personal services for adult males, and specified, usually low-prestige forms of productive labour (Brewer et al., 2002; Power, 2004).

Eco-feminists on the other hand have shown that there is a profound connection between the dismissal of housework, the devaluation of nature, and the idealisation of what is produced by technology and in the market. Alongside Agarwal's work in India, other contributions have focused on gendered aspects of colonialism (Perkins et al., 2005), land-grabs in Africa and how Nigerian women occupied oil terminals and inspired global protests against war and oil companies (Turner and Brownhill, 2004); or women's local grassroots organizing around environmental issues at the global level (Shiva, 2010).

The theme of care has been touched more tangentially, for example in the context of discussing land rights of rural class and gender (Cliffe and Moorsom, 1979; Nightingale, 2011) and feminization of anti-poverty programs in Latin America (Bradshaw et al., 2018). Dengler and Strunk (2018) note a potential connection to the literature on Degrowth here, as Degrowth is often motivated in terms of an alleviation of environmental injustices and a promotion of care and reproductive activities. However, the authors argue how this is possible only if degrowth work-sharing proposal if designed in a gender-sensitive way.

A new research agenda for Ecological Economics of Degrowth therefore needs to not only have a strong engagement with feminist debates, but also reflect on the gendered genealogies of knowledge produced and reproduced. The following section focuses on aspects for an emerging degrowth paradigm situated at the intersection of decolonial feminist research and knowledge production and proposes three research themes.

4.2.2. Research topic 5: ecological requirements of care work

A considerable amount of total human labour, up to one half in some countries is dedicated to unpaid care work (Benería, 2015), typically within the realm of the household and done mostly by women, many of them immigrant care workers. The concept of *embodied debt* refers to the unpaid reproductive labour that provides use value and regenerates the conditions of production, including that of future labour force (Salleh, 2009, 2018).

According to Čakardić (2017:42), the feminism of Rosa Luxemburg's work in *The Accumulation of Capital* can be a productive starting point to analyse embodied debt. The author argues that Luxemburg's theory "widely contributed to drawing a distinction between productive and non-productive labour. One such example comes from her interpretation of the societal role of the family". Čakardić (2017:44) furthermore shows how "Luxemburg differentiated between labour in the market sphere and labour in the household sphere, thereby laying the foundations for early social-reproduction theory. Luxemburg also argued that whenever capitalism is in crisis, or needs 'allies' for the restoration of profitability, it integrates, often legally, marginalised "Others" – women, children, non-white races – into the commodified sphere of accumulation".

Ecological economists have done some preliminary metabolic analyses showing how the commodification of care work - that is the substitution of unpaid care work with equivalent market services increases energy requirements (e.g. D'Alisa and Cattaneo, 2013). The commodification of care work therefore increases total energy use and goes against the objective of degrowing energy use. Ecological economists have the tools to develop an important research agenda of calculating in non-monetary/biophysical/material terms the actual dedication of human time to care labour, and to help us understand better the role of care work in social metabolism. From a degrowth perspective it is important to start thinking how systems of care provisioning could be reconfigured so as to enable both a reduction of throughput and more equitable allocation of the work along lines of gender, ethnicity, class or race. What are, for example, the metabolic requirements of a public care system compared to a private one? How does co-parenting or childcare sharing change the throughput of the services involved?

4.2.3. Research topic 6: role of women in ecological distribution conflicts and opposition to growth projects

Ecological economists have contributed significantly to our understanding of ecological distribution or environmental justice (EJ) conflicts (Martinez-Alier et al., 2016; Martínez-Alier et al., 2010; Pérez-Rincón et al., 2018). These contributions have promoted also understandings of links between such conflicts and the growth of the global economy or what Brand and Wissen (2018) call the imperial mode of living. The Environmental Justice Atlas registers a growing number of environmental justice movements, emerging as a result of geographically uneven and socially unequal metabolic processes, and many forms of hierarchy and exploitation (Temper et al., 2018). These conflicts, known as ecological distribution conflicts arise from social asymmetries in the distribution of political, economic and social power (O'Connor and Martinez-Alier, 1998). However, less attention has been paid to the gender dimensions of these conflicts (for an exception, see: Veuthey and Gerber, 2012).

Women figure at the forefront of many mobilizations against extractive projects. A new research agenda could focus on better understanding why and how. Research here should move beyond the early essentialist connection between women and nature that has been criticized by many ecofeminists (e.g. ideas that women defend nature because their reproductive function brings them closer to nature, or because they care more for the health of their children). It should instead shed light on the changing position of women in the context of changing metabolic flows and their related political economies. For example, why do women increasingly lead movements against mining? How does such participation reconfingure women's positions in their communities? How do claims about territoriality or new models of motherhood become part and parcel of such conflicts?

4.2.4. Research topic 7: gender relations in grassroots economies

In the degrowth literature, there is considerable emphasis on alternative economic practices – from urban gardens and time banks to alternative food networks and cooperatives. These are thought to be the seeds or incubators of alternative economic ethics and practices and hence symbolize a degrowth transition (see Kallis et al., 2018). There is however surprisingly little work on gender relations and the distribution of care-work within such initiatives.

A gendered division of time and tasks in the "mainstream economy" tends to reproduce itself in the alternative sphere, despite greater awareness for the need to dismantle traditional gender roles. Why and how does this happen? Case studies on the unequal division of labour within alternative or sharing economy experiments (e.g. Frenken and Schor, 2017; Schor et al., 2016) can shed light on the potential for reproduction of patriarchal relations, the mechanisms and causes of it as well as ways to avoid them even in a hypothetical post-growth or degrowth society. Research in this domain could – and should – be "Action Research" where researchers directly contributing to the alternative collectives they study and help them through the production of knowledge that is gender and power conscious.

5. Conclusions

Research on degrowth is an important domain of ecological economics. While existing research has focussed mostly on proving the limits to (green) growth and pondering economic alternatives, this article has argued that future research should pay more attention, and be written, from the 'margins' – from the perspective, that is, of the marginalized, in terms of gender, race, class and caste. Ecological economists can do much better in shedding light on the structure and causes of unequal North-South relations, or the ways women continue to be exploited within the growth economy, and the ways they challenge this exploitation. The state of the current ecological economic scholarship on these questions remain unsatisfactory. There is a need to go beyond the mere recognition of the important – and unpaid – care labour of women, or facile statements according to which the North should degrow, so as to exploit the South less (or let it grow).

In this review, we have identified a range of new questions the next generation of ecological economists can and should grapple with: what sort of post-growth policies can work in the Global South, and under what conditions? Who enjoys an imperial mode of living, where and what are the ecological economic processes of unequal exchange that sustain it? How do the concepts that ecological economists use - or produce - subtly or unintentionally reproduce colonial or gendered ways of seeing? How does the arrival of 'development' in peripheral areas change local metabolisms, and how do clashes between 'imperial' and indigenous social metabolisms play out in the ground? What are the metabolic requirements of different arrangements of care-work and how are they distributed along lines of class, race, ethnicity, and gender? Why are women increasingly leading environmental justice movements in contexts of ecological distribution conflicts, and how does their leadership reconfigure the content, framing and strategies of such mobilizations? What is the gender division of work in alternative, solidarity economies and how can we avoid a reproduction of unequal gender relations in a hypothetical degrowth transition?

Crucially, this research agenda should centre the voices of those writing from the margin; scholars from non-privileged positions in dominant academia. A new research agenda should integrate a plurality of genders, ethnicities, cultural and geographical backgrounds.

Acknowledgements

We are grateful to the two anonymous referees for their valuable knowledge and the time they put in reviewing this paper. We would also like to thank the editor Joshua Farley for his valuable comments and encouragement. A special thanks to Lena Weber for proofreading the earlier drafts of the manuscript. We are also very grateful to Joan Martinez-Alier for his invaluable insights and his constant support during the writing process. The research benefited from financial support from the Spanish Ministry of Economy and Competitiveness (MINECO), through the "María de Maeztu" program for Units of Excellence (MDM-2015-0552). Kallis' research benefited also from support from the COSMOS (CSO2017-88212-R) grant, also by MINECO. Avila, Roy and Hanacek Roy would like to acknowledge the support from the European Research Council (ERC) Advanced Grant ENVJUSTICE (No. 695446). Avila's research was also financially supported by CONACYT-Postgraduate Research Grant.

Appendix A. Supplementary data

Supplementary material related to this article can be found, in the online version, at doi:https://doi.org/10.1016/j.ecolecon.2019. 106495.

References

- Acosta, A., 2012. The Buen vivir an opportunity to imagine another world. In: Foundation, H.B. (Ed.), Inside a Champion. An Analysis of the Brazilian Development Model. Heinrich-Böll-Stiftung (Office Brazil). Glória, Rio de Janeiro, RJ, Brasil, pp. 192–210.
- Acosta, A., 2015. El Buen Vivir como alternativa al desarrollo. Algunas reflexiones económicas y no tan económicas. Política y Soc. 52, 299–330. https://doi.org/10. 5209/rev_POSO.2015.v52.n2.45203.
- Agarwal, B., 2009. Rule making in community forestry institutions: The difference women make. Ecol. Econ. 68, 2296–2308. https://doi.org/10.1016/j.ecolecon.2009. 02.017.
- Agarwal, B., 2010. Gender and Green Governance: The Political Economy of Women'S Presence Within and Beyond Community Forestry. Oxford University Press, New Delhi and Oxford.
- Akbulut, B., Demaria, F., Gerber, J.-F., Martínez-Alier, J., 2019. Who promotes sustainability? Five theses on the relationships between the degrowth and the environmental justice movements. Ecol. Econ. 165, 106418. https://doi.org/10.1016/j.ecolecon. 2019.106418.
- Avila-Calero, S., Pérez-Rincón, M., 2018. Prefacio a la edición latinoamericana. In: D'Alisa, G., Demaria, F., Kallis, G. (Eds.), Decrecimiento: Vocabulario Para Una Nueva Era. Edición Latinoamericana. Icaria-Heinrich Boll Mexico.
- Azamar-Alonso, A., Carrillo-González, G., 2017. Extractivismo y deuda ecológica en América Latina. Luna Azul 400–418. https://doi.org/10.17151/luaz.2017.45.20.
- Bauhardt, C., 2014. Solutions to the crisis? The Green New Deal, degrowth, and the solidarity economy: alternatives to the capitalist growth economy from an ecofeminist economics perspective. Ecol. Econ. https://doi.org/10.1016/j.ecolecon.2014.03.015.
- Bayon, D., Flipo, F., Schneider, F., 2010. La décroissance. 10 questions pour comprendre et en débattre. La Découverte, Paris.
 Benería, L., 2015. Gender, Development and Globalization: Economics as If All People
- Mattered, 2nd ed. Routledge.
- Bertinat, P., 2013. Un nuevo modelo energético para la construcción del buen vivir, in: Alternativas Al Capitalismo/Colonialismo Del Siglo XXI. Abya-YalaFundación Rosa Luxemburg, Qito, Ecuador, pp. 161–188.
 Borrero Navia, J.M., 1994. La Deuda Ecológica: Testimonio de Una Reflexión, 1st ed. Cali
- Borrero Navia, J.M., 1994. La Deuda Ecológica: Testimonio de Una Reflexión, 1st ed. Cali Fundación para la Investigación y Protección del Medio Ambiente FIPMA, Cali, Colombia.
- Bradshaw, S., Chant, S., Linneker, B., 2018. Challenges and changes in gendered poverty: The feminization, De-feminization, and Re-feminization of poverty in latin America. Fem. Econ. 0, 1–26. https://doi.org/10.1080/13545701.2018.1529417. Brand, U., Wissen, M., 2018. Imperial mode of living. Krisis 75–78.
- Brewer, R.M., Conrad, C.A., King, M.C., 2002. The complexities and potential of theorizing gender, caste, race, and class. Fem. Econ. https://doi.org/10.1080/ 135457002200019038.
- Bryan, S., Hoffman, B., 2007. Transparency and Accountability In Africa'S Extractive Industries: The Role of the Legislature. National Democratic Institute for International Affairs (NDI), Washington D.C., USA.
- Čakardić, A., 2017. From theory of accumulation to social-reproduction theory: A case for luxemburgian feminism. Hist. Mater. 25, 37–64. https://doi.org/10.1163/1569206X-12341542.
- Cliffe, L., Moorsom, R., 1979. Rural class formation and ecological collapse in Botswana. Rev. Afr. Polit. Econ. https://doi.org/10.1080/03056247908703395.
- Cosme, I., Santos, R., O'Neill, D.W., 2017. Assessing the degrowth discourse: A review and analysis of academic degrowth policy proposals. J. Clean. Prod. 149, 321–334. https://doi.org/10.1016/j.jclepro.2017.02.016.
- D'Alisa, G., Cattaneo, C., 2013. Household work and energy consumption: a degrowth perspective. Catalonia's case study. J. Clean. Prod. 38, 71–79. https://doi.org/10. 1016/j.jclepro.2011.11.058.
- D'Alisa, G., Demaria, F., Kallis, G., 2015. Introduction to degrowth. In: D'Alisa, G., Demaria, F., Kallis, G. (Eds.), Degrowth A Vocabulary For a New Era. Routledge, New York and London. 2 Park Square, Milton Park, Abingdon, Oxon OX14 4RN.
- Daly, H., 2013. Ecological economics and sustainable development. Selected Essays of Herman Daly. Edward Elgar Publishing.https://doi.org/10.4337/9781847206947. 9 Dewey Court Northampton Massachusetts 01060 USA.
- de Sousa Santos, B., 2016. Epistemologies of the South and the future. From Eur. South. 1,

17-29. https://doi.org/10.1080/10570310802636334.

- Demaria, F., Kothari, A., 2017. The Post-development dictionary agenda: paths to the pluriverse. Third World Q. 38, 2588–2599. https://doi.org/10.1080/01436597. 2017.1350821.
- Demaria, F., Schneider, F., Sekulova, F., Martinez-Alier, J., 2013. What is degrowth? From an activist slogan to a social movement. Environ. Values 22, 191–215. https:// doi.org/10.3197/096327113X13581561725194.
- Demmer, U., Hummel, A., 2017. Degrowth, anthropology, and activist research: the ontological politics of science. J. Polit. Ecol. 4, 611–622.
- Dengler, C., Seebacher, L.M., 2019. What about the global South? Towards a feminist decolonial degrowth approach. Ecol. Econ. 157, 246–252. https://doi.org/10.1016/j. ecolecon.2018.11.019.
- Dengler, C., Strunk, B., 2018. The monetized economy versus care and the environment: degrowth perspectives on reconciling an antagonism. Fem. Econ. 24, 160–183. https://doi.org/10.1080/13545701.2017.1383620.

 Dilli, S., Carmichael, S.G., Rijpma, A., 2018. Introducing the historical gender equality index. Fem. Econ. 5701, 1–27. https://doi.org/10.1080/13545701.2018.1442582.
 Domazet, M., Ančić, B., 2019. Complementarity between the EJ movement and degrowth

- Domazet, M., Ančić, B., 2019. Complementarity between the EJ movement and degrowth on the European semiperiphery: an empirical study. Ecol. Econ. 157, 120–128. https://doi.org/10.1016/j.ecolecon.2018.11.006.
- Escobar, A., 2011. Encountering Development The Making and Unmaking of the Third World. Princeton University Press, Princeton, New Jersey USA.

Escobar, A., 2015. Degrowth, postdevelopment, and transitions: a preliminary con-

- versation. Sustain. Sci. 10, 451–462. https://doi.org/10.1007/s11625-015-0297-5.
 Esteva, G., 2010. Development. In: Sachs, W. (Ed.), The Development Dictionary: A Guide to Knowledge as Power. Zed Books, London-New York.
- Esteva, G., 2018. Autonomía en américa Latina. In: D'allisa, G., Demaria, F., Kallis, G. (Eds.), Decrecimiento: Vocabulario Para Una Nueva Era. Edición Latinoamericana. Icaria-Heinrich Boll Mexico.
- Federici, S., 2018a. Women, money and debt: notes for a feminist reappropriation movement. Aust. Fem. Stud. 33, 178–186. https://doi.org/10.1080/08164649.2018. 1517249
- Federici, S., 2018b. Marx and feminism. TripleC 16, 468–475. https://doi.org/10.31269/ triplec.v16i2.1004.
- Flipo, F., 2007. Voyage dans la galaxie décroissante. Mouvements 50, 143–151.

Frenken, K., Schor, J., 2017. Putting the sharing economy into perspective. Environ. Innov. Soc. Transitions 23, 3–10. https://doi.org/10.1016/j.eist.2017.01.003.

Galeano, E., 2004. Las Venas Abiertas de América Latina. Siglo Veintiuno Editores, S.a. de C.v., Mexico D.F.

García Linera, Á., 2012. Geopolitics of the Amazon Vice-President of the Plurinational State of Bolivia. Translated by Robert Fidler. Links Intenational J, Soialist Reneval.

Geijzendorffer, I.R., Cohen-Shacham, E., Cord, A.F., Cramer, W., Guerra, C., Martín-López, B., 2017. Ecosystem services in global sustainability policies. Environ. Sci. Policy 74, 40–48. https://doi.org/10.1016/j.envsci.2017.04.017.

Georgescu-Roegen, N., 1979. Demain la Décroissance: Entropie-écologie-économie, Preface and Translation by Ivo Rens and Jacques Grinevald. PierreMarcel Favre, Lausanne

- Gerber, J.-F., Raina, R.S., 2018. Post-growth in the global South? Some reflections from India and Bhutan. Ecol. Econ. 150, 353–358. https://doi.org/10.1016/j.ecolecon. 2018.02.020.
- Giampietro, M., Mayumi, K., Sorman, A.H., 2013. Energy Analysis for a Sustainable Future: Multi-Scale Integrated Analysis of Societal and Ecosystem Metabolism. Routledge Taylor & francis Group London, London UK.

Gibbs, G., 2004. Qualitative Data Analysis: Explorations With NVivo: Understanding Social Research. Open University Press, Edmunsbury UK.

- Goeminne, G., Paredis, E., 2010. The concept of ecological debt: some steps towards an enriched sustainability paradigm. Environ. Dev. Sustain. 12, 691–712. https://doi.org/10.1007/s10668-009-9219-y.
 Gonzalez-Martinez, A.C., Schandl, H., 2008. The biophysical perspective of a middle in-
- Gonzalez-Martinez, A.C., Schandl, H., 2008. The biophysical perspective of a middle income economy: material flows in Mexico. Ecol. Econ. 68, 317–327. https://doi.org/ 10.1016/j.ecolecon.2008.03.013.
- Govindu, V.M., Malghan, D., 2005. Building a creative freedom: JC kumarappa and his economic philosophy. Econ. Polit. Wkly. https://doi.org/10.2307/4417581.
- Govindu, V.M., Malghan, D., 2016. The Web of Freedom: J. C. KumArappa and Gandhi'S Struggle for Economic Justice. Oxford University Press.
- Grosfoguel, R., 2002. Colonial difference, geopolitics of knowledge, and global coloniality in the Modern/Colonial capitalist world-system. Utop. Think. 25, 203–224. https:// doi.org/10.2307/40241548.
- Grosfoguel, R., 2011. A decolonial approach to political-economy: transmodernity, border thinking and global coloniality. J. Peripher. Cult. Prod. Luso-Hispanic World 1. https://doi.org/10.2307/40241659.
- Gudynas, E., 2009. El buen vivir más allá del extractivismo. Desco 70-81.
- Gudynas, E., 2014. El Buen vivir repolitiza los debates sobre el desarrollo. Rev. Kavilando 6, 27–29.
- Gudynas, E., Acosta, A., 2011. La renovación de la crítica al desarrollo y el buen vivir como alternativa. Utopía Y Prax. Latinoam. 16, 71–83.
 Guerrero-Osorio, A., 2018. Comunalidad. In: D'Alisa, G., Demaria, F., Kallis, G. (Eds.),
- Guerrero-Osorio, A., 2018. Comunalidad. In: D'Alisa, G., Demaria, F., Kallis, G. (Eds.), Decrecimiento: Vocabulario Para Una Nueva Era. Edición Latinoamericana. Icaria-Heinrich Boll Mexico.
- Hanaček, K., Rodríguez-Labajos, B., 2018. Impacts of land-use and management changes on cultural agroecosystem services and environmental conflicts—A global review. Glob. Environ. Chang. 50, 41–59. https://doi.org/10.1016/j.gloenvcha.2018.02.016.
- Hornborg, A., Martinez-Alier, J., 2016. Ecologically unequal exchange and ecological debt. J. Polit. Ecol. 23, 328. https://doi.org/10.2458/v23i1.20220.
- Kallis, G., Martinez-Alier, J., Norgaard, R.B., 2009. Paper assets, real debts: an ecologicaleconomic exploration of the global economic crisis. Crit. Perspect. Int. Bus. 5, 14–25. https://doi.org/10.1108/17422040910938659.

Kallis, G., Kerschner, C., Martinez-Alier, J., 2012. The economics of degrowth. Ecol. Econ. 84, 172–180. https://doi.org/10.1016/j.ecolecon.2012.08.017.

Kallis, G., Gómez-Baggethun, E., Zografos, C., 2013. To value or not to value? That is not

the question. Ecol. Econ. 94, 97-105. https://doi.org/10.1016/j.ecolecon.2013.07. 002.

- Kallis, G., Kostakis, V., Lange, S., Muraca, B., Paulson, S., Schmelzer, M., 2018. Research on degrowth. Annu. Rev. Environ. Resour. 43, 291–316. https://doi.org/10.1146/ annurev-environ-102017-025941.
- Kerschner, C., 2010. Economic de-growth vs. Steady-state economy. J. Clean. Prod. 18, 544–551. https://doi.org/10.1016/j.jclepro.2009.10.019.
- Khatua, S., Willam, S., 2006. Ecological debt: A case study from Orissa, India. In: Peralta, A.K. (Ed.), The Peoples of the South Are the Creditors. pp. 125–165.
- Kull, C., Arnauld de Sartre, X., Castro-Larrañaga, M., 2015. The political ecology of ecosystem services. Geoforum 61, 122–134. https://doi.org/10.1016/j.geoforum. 2015.03.004.
- Latouche, S., 2009. Farewell to Growth. Polity Presshttps://doi.org/10.1111/j.1468-2257.2011.00571.x.
- Lozano, F.A., Lopez, M.J., 2013. Border enforcement and selection of Mexican immigrants in the United States. Fem. Econ. 19, 76–110. https://doi.org/10.1080/13545701. 2012.752313.
- MacGregor, S., 2017. Routledge handbook of gender and environment. Routledge handb. Gend. Environ. 1–519. https://doi.org/10.4324/9781315886572.
- Martinez-Alier, J., 2014. The environmentalism of the poor. Geoforum 54, 239–241. https://doi.org/10.1016/j.geoforum.2013.04.019.
- Martinez-Alier, J., 2015. Ecological economics. International Encyclopedia of the Social & Behavioral Sciences, second edition. https://doi.org/10.1016/B978-0-08-097086-8. 91008-0.
- Martinez-Alier, J., Temper, L., Del Bene, D., Scheidel, A., 2016. Is there a global environmental justice movement? J. Peasant. Stud. 43, 731–755. https://doi.org/10. 1080/03066150.2016.1141198.
- Martínez-Alier, J., 2002Ea. Ecological debt and property rights on carbon sinks and reservoirs. Cap.. Nat., Soc.. 13, 115–119. https://doi.org/10.1080/ 104557502101245404.

Martínez-Alier, J., 2002Eb. The ecological debt. Kurswechsel 4, 5-16.

- Martínez-Alier, J., Pascual, U., Vivien, F.-D., Zaccai, E., 2010. Sustainable de-growth: mapping the context, criticisms and future prospects of an emergent paradigm. Ecol. Econ. 69, 1741–1747. https://doi.org/10.1016/j.ecolecon.2010.04.017.
- Matthews, H.D., 2016. Quantifying historical carbon and climate debts among nations. Nat. Clim. Chang. 6, 60–64. https://doi.org/10.1038/nclimate2774.
- McAfee, 1999. Green, selling nature to save it? Biodiversity and the rise of developmentalism. Environ. Plan. D Soc. Sp. 17, 133–154.
 Mignolo, W., 2018. Pensamiento decolonial. In: D'Alisa, G., Demaria, F., Kallis, G. (Eds.),
- Mignolo, W., 2018. Pensamiento decolonial. In: D'Alisa, G., Demaria, F., Kallis, G. (Eds.), Decrecimiento: Vocabulario Para Una Nueva Era. Edición Latinoamericana. Icaria-Heinrich Boll Mexico, Cuidad de Mexico.
- Mingorría, S., Gamboa, G., Martín-López, B., Corbera, E., 2014. The oil palm boom: socioeconomic implications for Q'eqchi' households in the polochic valley. Guatemala. Environ. Dev. Sustain. 16, 841–871. https://doi.org/10.1007/s10668-014-9530-0.
- Nelson, J.A., Power, M., 2018. Ecology, sustainability, and care: developments in the Field. Fem. Econ. 24, 80–88. https://doi.org/10.1080/13545701.2018.1473914.
 Neumayer, E., 2000. In defence of historical accountability for greenhouse gas emissions.
- Ecol. Econ. 33, 185–192. https://doi.org/10.1016/S0921-8009(00)00135-X. Ngosso, T., 2016. Ecological Debt Versus Financial Debt In The African Context: The
- Critique Illuminating The Challenges, Opportunities and Promises of a Rising Africa. Cap. Nat. Soc. 13, 115–119. Nightingale, A.J., 2011. Bounding difference: intersectionality and the material produc-
- tion of gender, caste, class and environment in Nepal. Geoforum 42, 153–162. https://doi.org/10.1016/j.geoforum.2010.03.004.
- O'Connor, M., Martinez-Alier, J., 1998. Sustainable development: concepts, rationalities and strategies. In: Faucheux, S., O'Connor, M., Straaten, J. (Eds.), Ecological Distribution and Distributed Sustainability, Economy & Environment. Springer Netherlands, Dordrecht, pp. 33–56. https://doi.org/10.1007/978-94-017-3188-1.
- O'Neill, D.W., Fanning, A.L., Lamb, W.F., Steinberger, J.K., 2018. A good life for all within planetary boundaries. Nat. Sustain. 1, 88–95. https://doi.org/10.1038/ s41893-018-0021-4.
- Olmedo, R., 2010. La contraproductividad del crecimiento económico. Universidad Nacional Autónoma de México, 1st ed. Ciudad Universitaria, Delegación Coyoacán, C.P. 04510, México, D.R.
- Pérez-Rincón, M., Vargas-Morales, J., Crespo-Marín, Z., 2018. Trends in social metabolism and environmental conflicts in four andean countries from 1970 to 2013. Sustain. Sci. 13, 635–648. https://doi.org/10.1007/s11625-017-0510-9.
- Perkins, E., 1997. Women, ecology and economics: New models and theories. Ecol. Econ. 20, 105–106. https://doi.org/10.1016/S0921-8009(96)00459-4. Perkins, P.E., 2007. Feminist ecological economics and sustainability. J. Bioeconomics 9,
- Perkins, P.E., 2007. Feminist ecological economics and sustainability. J. Bioeconomics 9, 227–244. https://doi.org/10.1007/s10818-007-9028-z.
- Perkins, E., Kuiper, E., Quiroga-Martínez, R., Turner, T.E., Brownhill, L.S., Mellor, M., Todorova, Z., Jochimsen, M.A., McMahon, M., Perkins, P.E., Kuiper, E., Quiroga-Martínez, R., Turner, T.E., Brownhill, L.S., Mellor, M., Todorova, Z., Jochimsen, M.A., McMahon, M., 2005. Explorations feminist ecological economics. Fem. Econ. 11, 107–150. https://doi.org/10.1080/13545700500301494.
- Power, M., 2004. Social provisioning as a starting point for feminist economics. Fem. Econ. 10, 3–19. https://doi.org/10.1080/1354570042000267608.
- Rangan, H., Sivaramakrishnan, K., 2001. Modern forests: statemaking and environmental change in colonial Eastern India. Econ. Geogr. 77, 197. https://doi.org/10.2307/ 3594064.
- Ravera, F., Scheidel, A., Dell'Angelo, J., Gamboa, G., Serrano, T., Mingorría, S., Cabello, V., Arizpe, N., Ariza, P., 2014. Pathways of rural change: an integrated assessment of

metabolic patterns in emerging ruralities. Environ. Dev. Sustain. 16, 811–820. https://doi.org/10.1007/s10668-014-9534-9.

- Rodríguez-Labajos, B., Yánez, I., Bond, P., Greyl, L., Munguti, S., Ojo, G.U., Overbeek, W., 2019. Not so natural an Alliance? Degrowth and environmental justice movements in the global south. Ecol. Econ. 157, 175–184. https://doi.org/10.1016/j.ecolecon. 2018.11.007
- Røpke, I., 2004. The early history of modern ecological economics. Ecol. Econ. 50, 293–314. https://doi.org/10.1016/j.ecolecon.2004.02.012.
- Roncevic, B., 2002. Path from the (semi) periphery to the core: on the role of sociocultural factors. IES Proc. 1, 1–26.
- Russi, D., Gonzalez-Martinez, A.C., Silva-Macher, J.C., Giljum, S., Martínez-Alier, J., Vallejo, M.C., 2008. Material flows in Latin America: A comparative analysis of Chile, Ecuador, Mexico, and Peru, 1980-2000. J. Ind. Ecol. 12, 704–720. https://doi.org/ 10.1111/j.1530-9290.2008.00074.x.
- Salleh, A., 2009. Ecological debt: embodied debt. In: Salleh, A. (Ed.), Eco-Sufficiency and Global Justice: Women Write Political Ecology. Pluto Press, pp. 336.
- Salleh, A., 2018. Deeper than Deep Ecology: The Eco-Feminist Connection, in: Feminist Ecologies. Springer International Publishing, Cham, pp. 25–33. https://doi.org/10. 1007/978-3-319-64385-4_2.
- Schor, J.B., Fitzmaurice, C., Carfagna, L.B., Attwood-Charles, W., Poteat, E.D., 2016. Paradoxes of openness and distinction in the sharing economy. Poetics 54, 66–81. https://doi.org/10.1016/j.poetic.2015.11.001.
- Sekulova, F., Kallis, G., Rodríguez-Labajos, B., Schneider, F., 2013. Degrowth: from theory to practice. J. Clean. Prod. 38, 1–6. https://doi.org/10.1016/j.jclepro.2012. 06.022.
- Shiva, V., 2010. Staying Alive: Women, Ecology and Development, 10th ed. South End Press.
- Siccama, C.J., Penna, S., 2008. Enhancing validity of a qualitative dissertation research study by using NVIVO. Qual. Res. J. 8, 91–103. https://doi.org/10.3316/ ORJ0802091.
- Silva-Macher, J.C., Farrell, K.N., 2014. The flow/fund model of conga: exploring the anatomy of environmental conflicts at the Andes-amazon commodity frontier. Environ. Dev. Sustain. 16, 747–768. https://doi.org/10.1007/s10668-013-9488-3.
- Sovacool, B.K., Walter, G., Van de Graaf, T., Andrews, N., 2016. Energy governance, transnational rules, and the Resource curse: exploring the effectiveness of the extractive industries transparency initiative (EITI). World Dev. 83, 179–192. https:// doi.org/10.1016/j.worlddev.2016.01.021.
- Spash, C.L., 2011. Social ecological economics: understanding the past to See the future. Am. J. Econ. Sociol. 70, 340–375. https://doi.org/10.1111/j.1536-7150.2011. 00777.x.
- Spencer, P., Perkins, P.E., Erickson, J.D., 2018. Re-establishing justice as a pillar of ecological economics through feminist perspectives. Ecol. Econ. 152, 191–198. https://doi.org/10.1016/j.ecolecon.2018.05.022.
- Srinivasan, U.T., Carey, S.P., Hallstein, E., Higgins, P.A.T., Kerr, A.C., Koteen, L.E., Smith, A.B., Watson, R., Harte, J., Norgaard, R.B., 2008. The debt of nations and the distribution of ecological impacts from human activities. Proc. Natl. Acad. Sci. 105, 1768–1773. https://doi.org/10.1073/pnas.0709562104.Svampa, M., 2015. Commodities consensus: neoextractivism and enclosure of the com-
- Svampa, M., 2015. Commodities consensus: neoextractivism and enclosure of the commons in latin America. South. Atl. Q. 114, 65–82. https://doi.org/10.1215/ 00382876-2831290.
- Svampa, M., 2016. Debates Latinoamericanos. Indianismo, Desarrollo, Dependencia Y Populismo, 1st ed. Buenos Aires Edhasa, Buenos Aires.
- Temper, L., Del Bene, D., Martinez-Alier, J., 2018. Mapping the frontiers and front lines of global environmental justice: the EJAtlas. J. Polit. Ecol. 22, 255. https://doi.org/10. 2458/v22i1.21108.
- Tortosa, J.M., 2016. Maldesarrollo y mal vivir. Pobreza y violencia a escala mundial. POLIS 28, 1–5.
- Turner, T.E., Brownhill, L.S., 2004. Why women are at war with chevron: Nigerian subsistence struggles against the International oil industry. J. Asian Afr. Stud. 39, 63–93. https://doi.org/10.1177/0021909604048251.
- Vallejo, M.C., 2010. Biophysical structure of the Ecuadorian economy, foreign trade, and policy implications. Ecol. Econ. 70, 159–169. https://doi.org/10.1016/j.ecolecon. 2010.03.006.
- Velicu, I., 2019. De-growing environmental justice: reflections from anti-mining movements in Eastern Europe. Ecol. Econ. 159, 271–278. https://doi.org/10.1016/j. ecolecon.2019.01.021.
- Veuthey, S., Gerber, J.F., 2012. Accumulation by dispossession in coastal Ecuador: shrimp farming, local resistance and the gender structure of mobilizations. Glob. Environ. Chang. https://doi.org/10.1016/j.gloenycha.2011.10.010.
- Chang. https://doi.org/10.1016/j.gloenvcha.2011.10.010.
 Weiss, J.S., 2017. The role of ecological economics in latin American public policy: latin American consensus at the ISEE Washington meeting. Ecol. Econ. 138, 238–241. https://doi.org/10.1016/j.ecolecon.2017.03.041.
- Weiss, M., Cattaneo, C., 2017. Degrowth taking stock and reviewing an emerging academic paradigm. Ecol. Econ. 137, 220–230. https://doi.org/10.1016/j.ecolecon. 2017.01.014.
- Woolley, F., 2005. The citation impact of feminist economics. Fem. Econ. 11, 85–106. https://doi.org/10.1080/13545700500301312.
- Xue, J., Arler, F., Næss, P., 2012. Is the degrowth debate relevant to China? Environ. Dev. Sustain. 14, 85–109. https://doi.org/10.1007/s10668-011-9310-z.
- Zibechi, R., 2006. Movimientos Sociales:Nuevos Escenarios Y Desafíos Inéditos. OSAL CLACSO.
- Zografos, C., Howarth, R.B., 2010. Deliberative ecological economics for sustainability governance. Sustainability 2, 3399–3417. https://doi.org/10.3390/su2113399.